Author's response to reviews

Title: Anterolateral approach with tibial tubercle osteotomy versus standard medial approach for primary total knee arthroplasty: does it matter?

Authors:

  Michael T Hirschmann (Michael.Hirschmann@ksbh.ch)
  Mathias Hoffmann (mathias.hoffmann@ksl.ch)
  Robert Krause (robert.krause@oberlinhaus.de)
  Reza-Amir Jenabzadeh (reza.jenabzadeh@imperial.ac.uk)
  Markus Arnold (Markus.Arnold@hirslanden.ch)
  Niklaus F Friederich (niklaus-f.friederich@unibas.ch)

Version: 2 Date: 17 September 2009

Author's response to reviews:

Thank you very much for your helpful feedback and suggestions. We made the following changes in response to your suggestions:

Accepted all minor changes. Added and deleted references in background section.

Background section- changed the first sentence. Methods section- moved part to results section. Clarified the follow-up part in terms of outcome measurements and instruments used. Also clarified statistical part. Rearranged the results section with regards to the methods.